Author's response to reviews

Title: Deletion of REXO1L1 locus in a patient with malabsorption syndrome, growth retardation, and dysmorphic features: a novel recognizable microdeletion syndrome?

Authors:

Maria Rosaria D’Apice (d.apice@med.uniroma2.it)
Antonio Novelli (a.novelli@css-mendel.it)
Alessandra di Masi (dimasi@uniroma3.it)
Michela Biancolella (michela.biancolella@uniroma2.it)
Antonio Antoccia (antoccia@uniroma3.it)
Francesca Gullotta (francescagullotta@hotmail.com)
Norma Licata (normalicata@hotmail.com)
Daniela Minella (danymin@gmail.com)
Barbara Testa (testabarbara@libero.it)
Anna Maria Nardone (annamaria.nardone@ptvonline.it)
Giampiero Palmieri (giampiero.palmieri@uniroma2.it)
Emma Calabrese (clbmme00@uniroma2.it)
Livia Biancone (biancone@med.uniroma2.it)
Caterina Tanzarella (tanzarel@uniroma3.it)
Marina Frontali (marina.frontali@ift.cnr.it)
Federica Sangiuolo (sangiuolo@med.uniroma2.it)
Giuseppe Novelli (novelli@med.uniroma2.it)
Francesco Pallone (pallone@med.uniroma2.it)

Version: 2
Date: 12 September 2014

Author's response to reviews:

Dear Editor,

please, find attached our manuscript entitled “Deletion of REXO1L1 locus in a patient with malabsorption syndrome, growth retardation, and dysmorphic features: a novel recognizable microdeletion syndrome?” that we would like to submit to the BMC Medical Genetics as Case Report.

In our study we analyzed the clinical features and the genetic status of a patient with gastrointestinal alterations and dismorphisms associated to new genomic deletion.

We upload a new version without the additional file about the patient’s photo because permission to publish the patient’s photo was not granted.

My Colleague and I will be indebted to you if this manuscript is considered for publication in the BMC Medical Genetics as Case Report.
Best Regards

Maria Rosaria D'Apice